Author’s response to reviews
Title: Medical expenses of urban Chinese patients with stomach cancer during 2002-2011: a hospital-based multicenter retrospective study

Authors:
Xiao-Jie Sun (xiaojiesun@sdu.edu.cn)
Ju-Fang Shi (shijf@cicams.ac.cn)
Lan-Wei Guo (guolanwei1019@126.com)
Hui-Yao Huang (huanghy314@sina.cn)
Neng-Liang Yao (ny5t@eservices.virginia.edu)
Ji-Yong Gong (269176488@qq.com)
Ya-Wen Sun (sunyawen0313@126.com)
Guo-Xiang Liu (lgx6301@163.com)
A-Yan Mao (myan77@126.com)
Xian-Zhen Liao (125844068@qq.com)
Ya-Na Bai (baiyana@lzu.edu.cn)
Jian-Song Ren (renjiansong@sina.com)
Xin-Yu Zhu (402198244@qq.com)
Jin-Yi Zhou (zhoujinyi74@sina.com)
Ling Mai (mailing65588018@163.com)
Bing-Bing Song (bingbingsong2008@126.com)
Yu-Qin Liu (liuyq970930@126.com)
Lin Zhu (zhulintt2002@126.com)
Ling-Bin Du (yjsdlb0407@126.com)
Qi Zhou (qizhou9128@163.com)
Xiao-Jing Xing (13940066477@163.com)
Pei-An Lou (lpa82835415@126.com)
Xiao-Hua Sun (nbsunxh@126.com)
Xiao Qi (13663350353@163.com)
Yuan-Zheng Wang (klzywwyz@126.com)
Rong Cao (wsjjxpj@163.com)
Author’s response to reviews:

RE: BMC Cancer - BCAN-D-17-01441R3

Medical expenses of urban Chinese patients with stomach cancer during 2002-2011: a hospital-based multicenter retrospective study

Dear Dr. Hamashima,

We appreciate your thorough review of our paper. We have addressed the comments in the new revision.

We use your letter as a template for our reply with our responses in italics.

Responses to the editor’s comments:

Comment #1: The authors responded correctly, but some corrections were not adopt in their text. Inappropriate spacing remains as follows; Page 1 Line 16 and 24, Page 2 Line19, Page 5 Line 25, Page 8 Line 29, and Page 10 Line 5. Please check the whole text more carefully.
Our response: Thanks for the editor’s suggestion. We have checked the whole text more carefully, and corrected all the above spacing errors and some other spacing errors the editor didn’t mention.

Comment #2: Although the authors explain the cited page of Reference No. 35, there is no cited page on Page 14 Lines 35-36. These corrections are basic requirement.

Our response: Revised accordingly.